Author's response to reviews

Title: Maternal Mortality at Muhimbili National Hospital in Dar-es-Salaam, Tanzania

Authors:

Andrea B Pembe (andreapembe@yahoo.co.uk)
Chetto Paulo (chettop@yahoo.com)
Brenda S D'Mello (Brenda.dmello@ccbrt.or.tz)
Jos van Roosmalen (J.J.M.van_Roosmalen@lumc.nl)

Version: 2  Date: 20 May 2014

Author's response to reviews:

Thanks for the comments. The responses are provided below:

Comment: Insert line and page numbers. Remove page breaks.
Response: Line numbers and page numbers inserted and the page breaks have been removed from the manuscript.

Kind regards,

Andrea B Pembe - Corresponding author